Author's response to reviews

**Title:** Design and Organization of the Dexamethasone, Light Anesthesia and Tight Glucose Control (DeLiT) Trial: a factorial trial evaluating the effects of corticosteroids, glucose control, and depth-of-anesthesia on perioperative inflammation and morbidity from major non-cardiac surgery

**Authors:**

- Basem Abdelmalak (abdelmb@ccf.org)
- Ankit Maheshwari (mahesha@ccf.org)
- Edward Mascha (maschae@ccf.org)
- Sunita Srivastava (srivass@ccf.org)
- Theodore Marks (markst1@ccf.org)
- WH Wilson Tang (tangw@ccf.org)
- Andrea Kurz (kurza@ccf.org)
- Daniel I Sessler (sessled@ccf.org)

**Version:** 8  **Date:** 9 June 2010

**Author's response to reviews:** see over
Response to Reviewers for MS: 1543005264354808

Design and Organization of the Dexamethasone, Light Anesthesia and Tight Glucose Control (DeLiT) Trial: A Factorial Trial Evaluating the Effects of Corticosteroids, Glucose Control, and Depth-of-Anesthesia on Perioperative Inflammation and Morbidity From Major Non-cardiac Surgery

Basem Abdelmalak, Ankit Maheshwari, Edward Mascha, Sunita Srivastava, Theodore Marks, WH Wilson Tang, Andrea Kurz and Daniel I Sessler

Dear BMC Editorial Production Team,

Per your request, we have made the following changes to our manuscript.

Formatting changes to be made:

| Major revisions (we require the author to make these changes) |

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods/Design, Discussion, Trial registration (if appropriate) Please remember to also update the abstract details on the submission page.

This has been reformatted to reflect the headings listed above.

Authors' contributions: Please use initials to refer to each author

This has been completed.

Tables: please note that we are unable to display text vertically within the cells: please re-layout your table without these vertical elements.

The vertical text in Table 3 has been changed